Author's response to reviews

Title: The efficacy of a comprehensive lifestyle modification programme based on yoga in the management of bronchial asthma: a randomized controlled trial

Authors:

Ramaprabhu Vempati (rpvempati@gmail.com)
Deepak KK (kkdeepak@gmail.com)
Bijlani RL (rambij@gmail.com)

Version: 2 Date: 18 August 2008

Author's response to reviews:

The Editorial Board has made the following comments:
"Small numbers were studied and the manuscript is judged to be low priority for Respiratory Research"

Clarifications:
This study was done based on a sample size which is statistically calculated by taking the most relevant and important variables in the study. However, owing to the high variability of some mediators (ECP, PGD2 metabolite) in the study, although there was a positive trend observed in these parameters, we could not make any firm conclusions based on our results.